Author's response to reviews

Title: C-reactive Protein in Aortic Valve Disease

Authors:

Pedro L Sanchez (pedrolsanchez@secardiologia.es)
AnnaMaria Mazzone (mazzone@ifc.cnr.it)

Version: 2 Date: 7 February 2006

Author's response to reviews:

Dear editor.

Thank you very much for your promptness and thorough review of our manuscript. The comments and suggestions made by the reviewers have been very useful. Enclosed you will find a de novo submission of the review now entitled "C-reactive protein in degenerative aortic stenosis"

In the novo submission we have considered all suggestions made by the reviewers.